Author’s response to reviews

Title: STUDY PROTOCOL FOR A PHASE II DOSE EVALUATION RANDOMIZED CONTROLLED TRIAL OF CHOLECALCIFEROL IN CRITICALLY ILL CHILDREN WITH VITAMIN D DEFICIENCY (VITDAL-PICU STUDY)

Authors:

James McNally (dmcnally@cheo.on.ca)
Karin Amrein (karin.amrein@yahoo.de)
Katharine O’Hearn (kohearn@cheo.on.ca)
Dean Fergusson (dafergusson@ohri.ca)
Pavel Geier (pgeier@cheo.on.ca)
Matthew Henderson (mhenderson@cheo.on.ca)
Ali Khamessan (akhamessan@euro-pharm.com)
Margaret Lawson (lawson@cheo.on.ca)
Lauralyn McIntyre (lmcintyre@ohri.ca)
Stephanie Redpath (sredpath@cheo.on.ca)
Hope Weiler (hope.weiler@mcgill.ca)
Kusum Menon (menon@cheo.on.ca)

Version: 2 Date: 06 Sep 2017

Author’s response to reviews:

Dear Drs. Lancaster and Thabane,

Thank-you for the reviewer comments, and for the opportunity to revise our manuscript. We have responded to the journal comments below.

p10 "difference" (in new text) should be replaced with "different"
RESPONSE: This has been corrected.

p14 "We will consider these criteria acceptable if was are able..." Change "was" to "we".
RESPONSE: This has been corrected.

p19 Correct spelling of Whitney (currently "Whiteny") and add "test" after "Chi-square".
RESPONSE: This has been corrected.